23rd January 2024

Dear reviewers,

Thank you for your comments regarding our manuscript entitled ‘Using Clinical Cases To Guide Healthcare’.

Your review points are all addressed below:

(1) Please list all author and institutional information in order.
   a. This has been amended in the manuscript

(2) Please add the Core tip section. The number of words should be controlled between 50-100 words.
   a. This has been amended in the manuscript

(3) Please provide 4-10 keywords.
   a. This has been amended in the manuscript

(4) Please add the author’s contribution section. The format of this section will be as follows: Author contributions: Wang CL, Liang L, Fu JF, Zou CC, Hong F and Wu XM designed the research; Wang CL, Zou CC, Hong F and Wu XM performed the research; Xue JZ and Lu JR contributed new reagents/analytic tools; Wang CL, Liang L and Fu JF analyzed the data; Wang CL, Liang L and Fu JF wrote the paper.
   a. This has been amended in the manuscript

(5) Please provide the PubMed numbers (https://pubmed.ncbi.nlm.nih.gov/) and DOI citation numbers (https://doi.crossref.org/simpleTextQuery) to the reference list and list all authors of the references. If a reference has no PMID and DOI, please provide the source website address of this reference.
   a. This has been amended in the references section

(6) All references must be cited in order in the form of [1], [2], [3].
   a. This has been amended in the references section

(7) Please add the content of the abstract section. An informative, unstructured abstract of no less than 200 words should accompany each manuscript. Abbreviations must be defined upon first appearance in the Abstract. Do not use non-standard abbreviations, unless they appear at least two times in the text preceding the first usage/definition.
   a. This has been amended in the manuscript and the abstract shortened to required length.

(8) The author(s) must add a table/figure to the manuscript. There are no restrictions on the figures (color, B/W) and tables.
   a. This has been added to the manuscript and files uploaded

(9) The title cannot start with "the, a, an".
   a. This has been amended
We hope the new manuscript is deemed acceptable by yourselves for publication.

Yours sincerely,

Dr Michael Colwill